Author’s response to reviews

Title: Genetic study of common variants at the Apo E, Apo Al, Apo CIII, Apo B, lipoprotein lipase (LPL) and hepatic lipase (LIPC) genes and coronary artery disease (CAD): variation in LIPC gene associates with clinical outcomes in patients with established CAD

Authors:

Dr Marco G Baroni (marco.baroni@uniroma1.it)
Andrea Berni (andrea.berni@uniroma1.it)
Stefano Romeo (maktub76@hotmail.com)
Marcello Arca (marcelloarca@libero.it)
Tullio Tesorio (tultes@libero.it)
Giovanni Sorropago (giosorr@infinito.it)
Umberto Di Mario (umberto.dimario@uniroma1.it)
David J Galton (D.J.GALTON@mds.qmw.ac.uk)

Version: 3 Date: 8 Sep 2003

Dear Dr Veitch,

I have modified the abstract as requested, explaining the scientific evidences behind our study. In the new form it should comply with the guidelines for authors.

With best regards

Marco Baroni, MD PhD